Author's response to reviews

Title: Absent otoacoustic emissions predict otitis media in young Aboriginal children: a birth cohort study in Aboriginal and non-Aboriginal children in an arid zone of Western Australia

Authors:

Deborah Lehmann (deborahl@ichr.uwa.edu.au)
Sharon Weeks (mbra6582@bigpond.net.au)
Peter Jacoby (peterj@ichr.uwa.edu.au)
Dimity Elsbury (dimityelsbury@yahoo.com.au)
Janine Finucane (janie.finucane@health.wa.gov.au)
Annette Stokes (anne_t Stokes@hotmail.com)
Ruth Monck (clrmonck@emerge.net.au)
Harvey Coates (harveyc@cyllene.uwa.edu.au)
Kalgoorlie Otitis Media Research Project Team (KOMRP)
(deborahl@ichr.uwa.edu.au)

Version: 10 Date: 27 August 2008

Author's response to reviews:

label (age in months) added to x axis of table 3